Author's response to reviews

Title: Alcohol consumption and carotid artery structure in Korean adults aged 50 years and older

Authors:

Young-Hoon Lee (lyh8275@hanmail.net)
Min-Ho Shin (mhshinx@paran.com)
Sun-Seog Kweon (uijingogo@paran.com)
Sung-Woo Choi (jcsw74@hanmail.net)
Hye-Yeon Kim (dainghi@naver.com)
So-Yeon Ryu (cansry@chosun.com)
Bok-Hee Kim (kimbh@chosun.ac.kr)
Jung-Ae Rhee (jarhee@healthis.org)
Jin-Su Choi (jschoix@chonnam.ac.kr)

Version: 5 Date: 4 September 2009

Author's response to reviews: see over
Author's response to reviews

Title: Alcohol consumption and carotid artery structure: a population-based cross-sectional study of Korean adults aged 50 years and older

Authors:

Young-Hoon Lee (lyh8275@hanmail.net)
Min-Ho Shin (mhshinx@paran.com)
Sun-Seog Kweon (ujingogo@paran.com)
Sung-Woo Choi (icsw74@hanmail.net)
Hye-Yeon Kim (dainghi@naver.com)
So-Yeon Ryu (cansry@chosun.com)
Bok-Hee Kim (kimbh@chosun.ac.kr)
Jung-Ae Rhee (jarhee@healthis.org)
Jin-Su Choi (jschoix@chonnam.ac.kr)

Version: 3 Date: 4 September 2009

Author's response to reviews: see over
Dear BioMed Central Editorial Team

We are pleased to hear from you that our manuscript has been accepted for publication. Thank you very much.
The authors would like to thank the reviewers for their comments that help improve the manuscript.
Enclosed with this letter please find a point-by-point response to concerns.
We hope that it is suitable to be reviewed and considered for publication. Please.
Thank you for your consideration.

Sincerely yours,

Young-Hoon Lee,
Min-Ho Shin,
Sun-Seog Kweon,
Sung-Woo Choi,
Hye-Yeon Kim,
So-Yeon Ryu,
Bok-Hee Kim,
Jung-Ae Rhee,
Jin-Su Choi

P.S. We need a simple certificate (letter) to be published. Would you please send us the certification?
Reviewer: Ulf Schminke

Reviewer's report:
The authors have adequately responded to the reviewer's comments and they have improved their manuscript significantly. There are 3 minor essential revisions:

Page 8, line 3: 'still' should be replaced by 'frozen'

Authors’ comments: We have corrected this.

Page 8, last line: 'cardiac' should be replaced by 'carotid'

Authors’ comments: We have corrected this.

Page 15, line 5: 'Finna' should be replaced by 'Finns'

Authors’ comments: We have corrected this.